Author's response to reviews

Title: Cycling and bone health: a systematic review.

Authors:

Hugo Olmedillas (hugo.olmedillas@gmail.com)
Alejandro González-Agüero (alexgonz@unizar.es)
Luis A Moreno (lmoreno@unizar.es)
José A Casajús (joseant@unizar.es)
Germán Vicente-Rodríguez (gervicen@unizar.es)

Version: 9 Date: 24 August 2012

Author's response to reviews: see over
Cover letter

Zaragoza, 24 August 2012

Dear Editor,

Please, find attached a new version of our manuscript entitled “Cycling and bone health: a systematic review.” to be reconsidered for publication in your prestigious Journal. Manuscript ID: 9115166637274345.

We would like to apologize for our mistake at submitting the new version of our manuscript. Now, we have deleted the table of contents from page 2, and highlighted in yellow all changes throughout the manuscript.

Sincerely yours

[Signature]

On behalf of all the co-authors.

Alejandro González-Agüero